Author's response to reviews

Title: Instrumented gait analysis and individually tailored interdisciplinary interventions for children with cerebral palsy: A randomised controlled trial protocol

Authors:

Helle M Rasmussen (helrasmussen@health.sdu.dk)
Niels W Pedersen (niels.w.pedersen@rsyd.dk)
Søren Overgaard (soeren.overgaard@rsyd.dk)
Lars K Hansen (lars.kjaersgaard.hansen@rsyd.dk)
Ulrike Dunkhase-Heinl (ulrike.dunkhase-heinl@rsyd.dk)
Yanko Petkov (yanko.petkov@rsyd.dk)
Vilhelm Engell (vilhelm.engell@rsyd.dk)
Richard Baker (r.j.baker@salford.ac.uk)
Anders Holsgaard-Larsen (ahlarsen@health.sdu.dk)

Version: 2 Date: 17 June 2014

Author's response to reviews:

Revision prior to editorial assessment

Point-by-point description of the changes made (17.06.2014):

1. Trial registration number

The text: Clinicaltrials.gov NCT02160457.

Have been changed to: Trial registration: ClinicalTrials.gov NCT02160457. Registered 6/6/2014.

2. Peer review of study protocols

The study has received funding from six state and regional governmental institutions as well as charitable foundations. The organizations did receive a summary version of the study protocol along with an application. However, the Organizations did not conduct a formal review of the entire study protocol.

We do not consider these approvals of the study as a formal scientific peer-reviewed. Thus, no letter or document, saying that this protocol has been peer-reviewed exist.

2.1 Ethical and Funding Approval Documentation

Ethical approval has been obtained by The Committee for Medical Research Ethics in the Region of Southern Denmark. The committee have reviewed the study protocol in order to secure that the project are undertaken such as to be justified with regard to research ethics.

All documents are in Danish. The work of translating these documents can be
initiated, if that I still find it necessary after the above explanations of their approvals. It is unlikely that the translations and the subsequent endorsement and signing by the all the institutions can be made within the next week.

2.2 Funding
The study has not undergone peer-review by the funding body (see the explanation above).

2.3 Study status
The study is on going and still collecting data.

3.4 Related Articles
No publications containing results of this study gave already been published or submitted to any journal.